

Name of Journal: *World Journal of Hepatology*

Manuscript NO: 33809

Manuscript Type: Review

Liver immunology and herbal treatment

Initially, we would like to thank to the Editors for allocating their time and effort.

All the corrections we would like to add and/or change are listed below

1. 'At page 17 and 18, in the "Herbal Treatments with Potential Therapeutic Effect on Liver", detailed information was added '.....'
2. At page 25 In the table 2 "immunestimulatory" was corrected as a "immunostimulatory".
3. At page 25 In the table 2 "CCl4" was corrected as a "CCl₄".
4. At page 19, "Figure about liver-immune system" was added.

Kind Regards

Prof. H Yasemin BALABAN